Author’s response to reviews

Title: Assessment of patient safety culture in primary health care in Muscat, Oman: A questionnaire-based survey

Authors:
Muna Habib AL Lawati (drmunali@gmail.com)
Stephanie Short (stephanie.short@sydney.edu.au)
Nadia Noor Abdulhadi (nadianoorah@yahoo.com)
Sathiya Murthi Panchatcharam (sathiya.m@omsb.org)
Sarah Dennis (sarah.dennis@sydney.edu.au)

Version: 3 Date: 21 Mar 2019

Author’s response to reviews:

Dear Editor BMC Family Practice,


Thank you for sending this manuscript for review. The reviewers’ comments have been very useful, and we have revised the manuscript accordingly. Our responses to the reviewers’ comments are detailed below and have been addressed in the manuscript.

1. While assessing your manuscript in-house, we found several instances where the text displayed similarities to text found in other previously published sources. While we understand that you may wish to express some of the same ideas contained in these publications, please be aware that we cannot condone the use of text from previously published work. We would therefore be grateful if you could reformulate the sections listed below to resolve the overlap between your manuscript and other sources. Please ensure that, where relevant, these sources are also referenced as appropriate:

   Background –

Thanks for highlighting this error, the section has been rewarded.

2. We notice numerous spelling errors throughout the manuscript. Please perform a thorough copy-edit of your manuscript to resolve these errors.

The Manuscript has been revised for spelling errors and re corrected.

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colors. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Clean version of the manuscript has been uploaded.